Author’s response to reviews

**Title:** Comparison of Videolaryngoscope-guided versus Standard Digital Insertion Techniques of the ProSealTM Laryngeal Mask Airway: A Prospective Randomized Study

**Authors:**

Ulku Ozgul (ulku.ozgul@inonu.edu.tr)

Feray Erdil (feray.erdil@inonu.edu.tr)

Mehmet Erdogan (drmalierdogan@gmail.com)

Zekine Begec (zekine.begec@inonu.edu.tr)

Cemil Colak (cemil.colak@inonu.edu.tr)

Aytac Yucel (aytac.yucel@inonu.edu.tr)

Mahmut Durmus (mahmut.durmus@inonu.edu.tr)

**Version:** 2  **Date:** 11 Dec 2019

**Author’s response to reviews:**

1. Reviewer comments

-- Please address the remaining comment from reviewer 3, seen below

The change which made according to comment of reviewer 3 has been added in the manuscript discussion section (page 14, line 7-11)

“At the suitable depth of anesthesia, spontaneous breathing may be easily inhibited by opioids and hypnotic drugs without neuromuscular blockers. No using neuromuscular blocking agent prevents the undesirable side effects of these agents, such as prolonged neuromuscular block, and may lead to the need of neuromuscular antagonist [9, 24]. So, we did not prefer a neuromuscular blocking agent.”

2- CONSORT

It was reported that this study was performed according to CONSORT guidelines in Methods section.

3- Trial registration
The name of the trial registry and the date of trial registration has been added at the end of Abstract.

4- All comments for the authors has been uploaded in this box.

Ida Di Giacinto (Reviewer 3)

Q1.

It's correct but it would have been desirable for you to explain: "why you didn't use the NMBAs?" about the use of neuromuscular blocking agents or not its use.

A1.

The change which made according to your comment has been added in the manuscript discussion section (page 14, line 7-11)

“At the suitable depth of anesthesia, spontaneous breathing may be easily inhibited by opioids and hypnotic drugs without neuromuscular blockers. No using neuromuscular blocking agent prevents the undesirable side effects of these agents, such as prolonged neuromuscular block, and may lead to the need of neuromuscular antagonist [9, 24]. So, we did not prefer a neuromuscular blocking agent.”